Author's response to reviews

Title: Systematic review and mixed treatment comparison of randomized clinical trials of primary oral antifungal prophylaxis in allogeneic hematopoietic cell transplant recipients

Authors:

Eric J Bow (EJBow001@shaw.ca)
David J Vanness (dvanness@wisc.edu)
Monica Slavin (Monica.Slavin@mh.org.au)
Catherine Cordonnier (carlcord@club-internet.fr)
Oliver A Cornely (oliver.cornely@zks-koeln.de)
David I Marks (David.Marks@UHBristol.nhs.uk)
Antonio Pagliuca (antonio.pagliuca@kcl.ac.uk)
Carlos Solano (solano_car@gva.es)
Lael Cragin (lael.cragin@unitedbiosource.com)
Alissa J Shaul (Alissa.Shaul@evidera.com)
Sonja Sorensen (Sonja.Sorensen@evidera.com)
Richard Chambers (Richard.Chambers@pfizer.com)
Michal Kantecki (Michal.Kantecki@Pfizer.com)
David Weinstein (David.Weinstein@pfizer.com)
Haran Schlamm (HTSchlamm@att.net)

Version: 2 Date: 8 April 2014

Author's response to reviews:

We have included author email addresses on the title page of the manuscript, a statement on ethics, and introduced line numbers throughout. Thank you.